Author’s response to reviews

Title: AN ATYPICAL AND BILATERAL PRESENTATION OF CHARCOT FOOT DISEASE

Authors:
Chariclia Loupa (chariclia.loupa@gmail.com)
E. Meimeti (e.meimeti@gmail.com)
A. Kokas (gram_dioikhshs@flemig-hospital.gr)
E.D. Voyatzoglou (podiatry21@gmail.com)
A. Donou (andriana.donou@yahoo.gr)

Version: 2 Date: 30 Jul 2019

Author’s response to reviews:

A table summarizing the main clinical and biochemical information is added.

A more complete medical history is added.

Changes (except Table) are highlighted in yellow.